Author's response to reviews

Title: Secular changes in height, weight and body mass index in Hong Kong Children

Authors:

Hung-kwan So (sohk@cuhk.edu.hk)
Edmund AS Nelson (tony-nelson@cuhk.edu.hk)
Albert M Li (albertmli@cuhk.edu.hk)
Eric MC Wong (mcwong@cuhk.edu.hk)
Joseph TF Lau (jlau@cuhk.edu.hk)
Georgia S Guldan (gsguldan@cuhk.edu.hk)
Kwok-hang Mak (drkhmak@dh.gov.hk)
Youfa Wang (ywang@jhsph.edu)
Tai-fai Fok (taifai@cuhk.edu.hk)
Rita YT Sung (yntzsung@cuhk.edu.hk)

Version: 5 Date: 16 September 2008

Author's response to reviews:

The following additional formatting changes were requested

Removal of figure title from figures. This has been done.

Figure cropping - The figures have revised to crop them as closely as possible to minimise white space around the image.

Table order - Currently Table 5 is referenced before table 4. This order has been reversed and the tables reordered and referenced.